Math 144 Fall 2006 R. Schultz

UPDATED GENERAL INFORMATION — DECEMBER 8, 2006

(Second posting for December 8)

Here are two additions to the previous remarks on the final exam. First, despite the remarks excluding Section VII.3 from the exam material, looking at Exercise VII.3.2 and its solution is recommended. Also, a little more than half the exam will involve proofs of some kind; some may be formal, and others (e.g., induction arguments and looking at specific examples) may be less so.

Availability next week

The discussion section leader (Mr. Morton) has indicated that he will be available Monday afternoon, and I will be available from 10 A.M. to 12 noon on Tuesday. I have reserved Surge 284, so if you have some last minute questions please come to that room during those hours.

I should have final grades available by the end of the week. If you would like to know your grade, probably the best way to get in touch with me is by electronic mail. You will be free to take the exam after grades have been determined, and I will keep them for at least the first few weeks of the Winter 2007 Quarter if you wish to pick them up.